Author's response to reviews

Title: Awareness, agreement, adoption and adherence to type 2 diabetes mellitus guidelines: a survey of Indonesian primary care physicians

Authors:
- Indah S Widyahening (indah_aribowo@yahoo.com)
- Yolanda van der Graaf (Y.vanderGraaf@umcutrecht.nl)
- Pradana Soewondo (soewondops@yahoo.com)
- Paul Glasziou (pglaszio@bond.edu.au)
- Geert JMG van der Heijden (Geert.vander.Heijden@acta.nl)

Version: 4 Date: 12 February 2014

Author's response to reviews: see over
To Editor

BMC Family Practice Journal

Dear Editor,

Thank you for further suggestion toward the improvement of manuscript entitled “Awareness, agreement, adoption and adherence of physicians to type 2 diabetes mellitus guidelines in Indonesia”. Please find enclosed our revised manuscript which we hope adequately address the comments.

Regarding your comment to expand the conclusion, we have revise it in to the following,

“Our study shows that high awareness of the guideline does not always lead to adoption nor adherence to its recommendations. The production and dissemination of guidelines alone is not sufficient to ensure that research evidence gets into practice. Improvement of clinicians’ awareness, agreement, and adoption of guidelines need to be incorporated in to strategies to improve guideline adherence.”

We hope that this manuscript could be further processed for publication in your journal.

Sincerely yours,

[Signature]

Indah S. Widyahening, MD
Faculty of Medicine University of Indonesia

Also on behalf of all the co-authors